\documentclass[12pt]{exam}
\usepackage{amsmath, amsthm, amssymb}
\usepackage{fullpage, paralist}
\usepackage{subfig}
\usepackage{graphicx}
\usepackage{enumerate}
\usepackage{enumitem}


\newcommand{\Z}{{\mathbb Z}}
\newcommand{\Q}{{\mathbb Q}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\C}{{\mathbb C}}

\newcommand{\A}{{\mathcal A}}
\newcommand{\I}{{\mathcal I}}


\newcommand{\ds}{\displaystyle}
\newcommand{\la}{\langle}
\newcommand{\ra}{\rangle}
\newtheorem{thm}{Theorem}[section]

\theoremstyle{remark}
\newtheorem*{nnremark}{Remark}

\printanswers
%\noprintanswers
\unframedsolutions


\begin{document}

\parskip=\medskipamount \thispagestyle{empty}


\centerline{Math 36 --- Daily Homework}

\centerline{Assigned: 9/24} 

\centerline{Due: 9/27}
  

\begin{questions}

\question 

We have two general methods of finding answers to questions:

\begin{itemize}
\item analytic mathematical techniques
\item computer simulations
\end{itemize}

Compare and contrast these techniques. What are the pros and cons for each? Which is more compelling? Which do you prefer?


\end{questions}


\end{document}







%%% Local Variables:
%%% mode: latex
%%% TeX-master: t
%%% End: